

The State-of-the-Art in Air Pollution Monitoring and Forecasting Systems Using IoT, Big Data, and Machine Learning

Amisha Gangwar^{1,2} • Sudhakar Singh¹ • Richa Mishra¹ • Shiv Prakash¹

Accepted: 25 February 2023 / Published online: 16 March 2023 © The Author(s), under exclusive licence to Springer Science+Business Media, LLC, part of Springer Nature 2023, corrected publication 2023

Abstract

The quality of air is closely linked with the life quality of humans, plantations, and wildlife. It needs to be monitored and preserved continuously. Transportations, industries, construction sites, generators, fireworks, and waste burning have a major percentage in degrading the air quality. These sources are required to be used in a safe and controlled manner. Using traditional laboratory analysis or installing bulk and expensive models every few miles is no longer efficient. Smart devices are needed for collecting and analyzing air data. The quality of air depends on various factors, including location, traffic, and time. Recent researches are using machine learning algorithms, big data technologies, and the Internet of Things to propose a stable and efficient model for the stated purpose. This review paper focuses on studying and compiling recent research in this field and emphasizes the Data sources, Monitoring, and Forecasting models. The main objective of this paper is to provide the astuteness of the researches happening to improve the various aspects of air polluting models. Further, it casts light on the various research issues and challenges also.

Keywords Air Pollution \cdot Monitoring \cdot Forecasting \cdot Internet of Things (IoT) \cdot Big Data \cdot Machine Learning (ML)

Sudhakar Singh sudhakar@allduniv.ac.in

Amisha Gangwar agangwar@syr.edu

Richa Mishra richa_mishra@allduniv.ac.in

Shiv Prakash shivprakash.cse@gmail.com

¹ Department of Electronics and Communication, University of Allahabad, Prayagraj, India

² School of Information Studies, Syracuse University, New York, US

1 Introduction

Air pollution is exacerbating life quality on the earth. This alarming situation needs proper solutions to monitor the air, locate the sources causing air pollution and limit the use of those sources. Outside from the parking lots to inside the houses, the source of air pollution is present everywhere. Nearly 4 million people die per year due to illness caused by inefficient techniques used for household cooking. Among these deaths, 27% are from pneumonia, the other 27% from ischaemic heart disease, 20% from COPD (chronic obstructive pulmonary disease), remaining 18% and 8% are stroke and lung cancer, respectively [1]. Household Air Quality (HAQ) depends mainly on the kind of cookstove used and the geometry of the kitchen. Classic traditional cookstoves must be replaced with clean cooking solutions like Induction stoves, Forced drafted cookstoves, and Traditional cookstoves with hood structures. These cooking solutions lower the concentrations of Black Carbon, Carbon Monoxide (CO), and other pollutants emitted during the cooking process [2]. This all is not just limited to household pollution. Ambient pollution has more worsen outcomes. Smog, the brownishyellow coloration in the atmosphere, is caused due to the presence of pollutants like ozone, nitrogen oxides, and sulfur oxides. When nitrogen oxides released from automobiles, factories, or power plants react with any volatile organic compound in the presence of sunlight results in photochemical smog in the atmosphere, the end products of these reactions are severely harmful to human health [3]. Analysis of health impacts in crowded urban areas of the Bangkok Metropolitan Region (BMR) was done through photochemical smog modeling of PM2.5, and studies showed that meeting the limits set by World Health Organisation (WHO) for PM_{2.5} can avoid 1,415 death annually in that region [4]. According to Iaccarino et al. [5], there is a strong connection between ambient pollution and loss of cognitive abilities in older adults, whereas continuous exposures to Carbon Monoxide coming from fuelburning can lead shortly to fatigue, drowsiness, weakness, and difficulty in breathing and later severely to diseases like lung cancer and brain cancer. Haemoglobin has more affinity to Carbon Monoxide than Oxygen. If a person gets exposed to Carbon Monoxide, blood haemoglobin will make a strong bond with it and hinder the supply of oxygen to other parts [6]. The presence of sulfur dioxides and nitrogen oxides after the combustion of fossil fuels causes acid rain during precipitation. This acid rain disrupts humans and even wildfire and vegetation [7]. Juzhang Ren [8] studies show that ozone pollution near winter wheat fields also affects the economic yield by a 2.8% loss. All these damages need to be appropriately scaled. Hence, an interesting value evaluation system was established by Dongyue Liu [9], which assesses the environmental damages due to air pollution and provides legal compensation for the individual or organization that caused the one. However, merely evaluation methods are insufficient to address these issues; an active mechanism is required to cease or control the pollution from all the primary sources. For example, a setup can be constructed to regulate the pollution within defined standards from industries, or these systems can be installed around the pulmonology department of the hospital. From a commercial standpoint, these systems come with great investment results as these systems can be used by every sector dealing with bad air quality.

To preserve the air quality, various researches are going around the problem, i.e., pollution detection, Monitoring, source detection, and forecasting. These help in planning calculative strategies towards the problem [10]. For example, in [11], a low-cost setup was proposed using the Internet of Things (IoT) for monitoring purposes, whereas a similar approach was



Fig. 2 IoT, Big Data, and ML-based Air Pollution Monitoring and Prediction System

also made in [12] to monitor the data and further extended to a routing and predicting system. Studies mentioned in [13–15] are extended to examine what factors affect the problem more closely, what algorithms may improvise the forecasting feature, how to place the IoT setups, and how stable and economical the models are in real life. So, there are many aspects of the problem that need to be considered for a better solution.

Advanced technologies are making their imprints in every field, providing a whole new dimension to every other area. Since the last few decades, Artificial Intelligence (AI), Machine Learning (ML) algorithms, IoT, and Big Data tools are overtaking all primitive methods, giving out cost-friendly and many promising results for the stated problem [16], [17]. Figure 1 shows how these advanced technologies are used in Monitoring and prediction system. Also, Fig. 2 illustrates a general model that should be installed at every air station to take calculative measures in that area to deplete the air pollution in that area. In this direction, the main objective of this paper is to provide an overview of all the comprehensive research and an insightful review of the current state-of-the-art regarding the problem and the significant issues and challenges for an effective and efficient system. The main contributions of this paper can be summarized as follows.

- Various criteria and standards set by different governments and agencies for Air Quality Index are investigated.
- A systematic review of IoT-based real-time air monitoring systems is carried out, and optimal deployment of sensors is also emphasized.
- A comparative analysis of machine learning over classical statistical forecasting methods has been done.

- A systematic review and critical analysis of big data and machine learning-based models and comparisons on the various parameters are carried out.
- Recent research issues and challenges for efficient air pollution monitoring and forecasting are highlighted.

This is a review paper for air pollution monitoring and forecasting techniques. It is initiated with Sect. 1, which is a general introduction about air pollution and its effects along with the objective of this study. Section 2 discusses various air pollution standards proposed by the authorities and researchers. Section 3 presents the IoT infrastructure, reviews the various IoT-based models, and discusses the optimal placement of IoT setups. Section 4 discusses machine learning over classical statistical forecasting methods, examines the different Big Data and ML-based models, and compares them on multiple parameters. Section 5 focuses on the research issues and challenges for an effective air pollution monitoring and forecasting system, and Sect. 6 concludes the paper with the direction of future works. Figure 3 schematically represents the structure of the paper after the introduction.

2 Air Quality Standards

Air pollution is one of the sustainable concerns that is also mentioned in Sustainable Development Goals [18]. It is explicitly mentioned in SDG 3.9, SDG 11.6 and indirectly linked with SDG targets [19]. Unsustainable growth in urbanization as well as in industrialization leads to major respiratory illness. The various governments and agencies have set the air quality standards, which are described as follows.

2.1 US Air Monitoring Criteria

Six pollutants are declared as criteria pollutants by U.S. EPA (Environmental Protection Agency)– Carbon Monoxide (CO), Lead (Pb), Nitrogen Dioxide (NO₂), Ozone (O₃), Particulate Matter (PM) or Particle Pollution, and Sulphur Dioxide (SO₂). There are two subcategories for Particulate Matter based on pollutant size – $PM_{2.5}$ and PM_{10} . EPA has also fixed the averaging time and levels for each of the aforementioned pollutants, as shown in



Fig. 3 Structure of the Paper

Table 1Criteria Pollutants toMeasure Pollution [20]					
	Pollutant	Primary Seconda	(P)/ ry (S)	Averaging Time	Level
	Carbon Monox-	Р		8 h	9 ppm
	ide (CO)			1 h	35 ppm
	Lead (Pb)	P and S		Rolling 3-month average	$0.15 \mu g/m^3$
	Nitrogen Diox-	Р		1 h	100 ppb
	ide (NO ₂)	P and S		One year	53 ppb
	Ozone (O_3)	P and S		8 h	0.070 ppm
	Particle Pollu-	PM _{2.5}	Р	1 year	$12.0 \mu g/m^3$
	tion (PM)		S	1 year	$15.0 \mu g/m^3$
			P and S	24 h	$35 \ \mu g/m^3$
		PM_{10}	P and S	24 h	$150 \ \mu g/m^3$
	Sulphur Diox-	Р		1 h	75 ppb
	ide (SO ₂)	S		3 h	0.5 ppm
Table 2 AQI Classification [23]	Range of AQI Va	lues		Levels of Health Co	oncern
	0-50			Good	
	51-100			Moderate	

	151-200	Unhealthy
	201-300	Very Unhealthy
	301-500	Hazardous
Table 1. Primary and s	secondary, mentioned in Ta	ble 1, are the state in which commodi-
ties are effective. Prim	ary mainly includes the h	ealth of senior citizens, infants, asthma
patients, whereas Secon	ndary consists of the health	of crops, animals, vegetation. The units
used for levels are ppn	n – parts per million by vol	ume, ppb – parts per billion by volume,
$\mu g/m^3 - micrograms per$	er cubic meter of air [20]. T	he Air Quality Index (AQI) is a measure
of how polluted the air	is. It is a piecewise linear fu	unction of air pollutant concentration. As

101-150

described in [21] and [22], the following formula is used to calculate AQI.

$$AQI = \frac{AQI_{high} - AQI_{low}}{PC_{high} - PC_{low}} (PC - PC_{low}) + AQI_{low}$$
(1)

where,

AQI = Air Quality Index,

PC = pollutant concentration,

 $PC_{low} = \text{concentration-breakpoint} \leq PC$,

 $PC_{high} = \text{concentration-breakpoint} \geq PC$,

 $AQI_{low} =$ index-breakpoint respective to PC_{low} ,

 AQI_{high} = index-breakpoint respective to PC_{high} .

After calculating the AQI value, it is referred to Table 2 to check the classification. Each country has its air quality indices classification. Table 2 is for the United States [23].

Unhealthy for Sensitive Groups

Table 3 EU Standards [25]	Qualitative name	Index or sub-index	Pollutant µg/m ³	(hourly) co	oncentratio	n in
			NO ₂	PM ₁₀	O ₃	PM _{2.5} (optional)
	Very low	0–25	0-50	0-25	0–60	0–15
	Low	25-50	50-100	25-50	60–120	15-30
	Medium	50-75	100-200	50–90	120-180	30–55
	High	75-100	200-400	90-180	180-240	55-110
	Very high	>100	>400	>180	>240	>110

Table 4 National Air Quality Index (NAQI) of India [29]

AQI Category (Range)	PM_{10}	PM_{25}	NO ₂	O ₃	CO	SO ₂	NH ₃	Pb
	(24 h)	(24 h)	(24 h)	(8 h)	(8 h)	(24 h)	(24 h)	(24 h)
Good (0–50)	0-50	0–30	0–40	0–50	0-1.0	0-40	0-200	0-0.5
Satisfactory (51-100)	51-100	31-60	41-80	51-100	1.1-2.0	41-80	201-400	0.5-1.0
Moderately polluted (101–200)	101–250	61–90	81–180	101–168	2.1–10	81–380	401-800	1.1-2.0
Poor (201–300)	251-350	91–120	181–280	169–208	10–17	381– 800	801– 1200	2.1-3.0
Very poor (301–400)	351-430	121– 250	281-400	209–748	17–34	801– 1600	1200– 1800	3.1– 3.5
Severe (401–500)	430+	250+	400+	748+	34+	1600 +	1800+	3.5+

2.2 European Air Monitoring Criteria

European Union (EU) has a whole extensive body of legislatures that monitor ambient air pollution. They have set the standards as well as the objective for each pollutant present in the air [24]. European Environment Agency in November 2017 declared the EAQI (European Air Quality Index), and since then, it is being encouraged [25]. Selected EU standards are summarised in Table 3, taken from [25].

2.3 Indian Air Monitoring Criteria

Out of the world's 20 most polluted cities, 13 cities are from India alone. Some of them are Kanpur, Faridabad, Gaya, Varanasi, Patna, Delhi, and Lucknow [26]. This poor situation of breathing environment needs the attention of Indian government authorities and the participation of citizens. Under Swachh Bharat Abhiyan in October 2014, the government announced National Air Quality Index [27]. Along with State and Central Pollution Control Boards, National Air Monitoring Programs (NAMP) operate in 300+cities and have 700+monitoring stations [28]. Table 4 shows the Indian Air Quality Index [29].

These criteria and standards are followed in most of the literature. For examples, Kok et al. [30] and Alaoui et al. [31] used criteria similar to the US, whereas Nashh et al. [32] mentioned the US and European Standards. European Standards is also used in Lazrak et al. [33]. Indian Standards are used in Moses et al. [12] Monitoring and Re-routing system.

3 Internet of things Based Models

3.1 Internet of Things (IoT)

It is a collection of smart embedded sensors, devices, and software that helps to collect data and exchange it over the internet [34]. It provides the system with real-time data. The IoT architecture consists of the perception layer, network layer, and third-party application or cloud layer [35].

i) Perception Layer

This layer is also called the sensing layer. It consists of sensors. Sensors sense the data and send it periodically to the gateway or the cloud [35]. So, these sensors measure the levels of the specific pollutant. There are particular kinds of sensors to sense the particular pollutant, shown in Fig. 4.

a) Metal Oxide Sensors

These are low-cost sensors with good sensitivity and long response times. These sensors are easily affected by temperature and humidity. It is usually used to measure O_3 , CO, and NO_2 [36].

b) Optical Particulate Counter

These are moderate-cost sensors with a fast response time and a sensitivity range of $1 \ \mu g/m^3$. It can measure the size of the particle. Hence, this is used for measuring PM_{2.5} and PM₁₀, and the output depends on various factors such as color and density, humidity, refractive index, and shape [36].

- c) Optical Sensors These are moderate-cost sensors used to measure CO₂ and CO. They have good sensitivity for CO₂ with a response time of 20–120 s [36].
- d) Electrochemical Sensors

These sensors are of moderate cost with good sensitivity (mg/m³ to μ g/m³) and quick response time. It is usually used for measuring NO₂, SO₂, O₃, NO, and CO [36].



Fig. 4 Various kinds of Sensors

Table 5 gives the names of some specific IoT sensors and their details like cost (in Indian rupees), usages, and range of detection. The costs mentioned here are taken form e-commerce portals.

ii) Network Layer

This layer is the bridge between perception and application layers. This layer consists of gateways. Gateways are local schedulers, regulators, and processors. They do perform light processing to decrease unnecessary transmission [35].

iii) Third-Party Application or Cloud Layer

This layer provides the user with an interface to work on the data. It stores and analyzes the data and provides the final result. It can also perform real-time processing and learn through many available ML algorithms [35]. Fig. 5 represents the flow of collected data through all layers.



Fig. 5 Flow of data in an IoT setup

3.2 IoT Based Air Monitoring Systems

The IoT-based air monitoring system provides real-time location data for mining purposes. To construct an IoT setup, various embedded sensors, smart objects, network/ internet, software, cloud services, and devices are structured and integrated to meet the stated goal. Fig. 6 depicts a general design of IoT based Air Monitoring Systems.

In Fig. 6, all the setup components are connected via Network/Internet providing communicative abilities to the model. IoT Sensors (like MQ9, MQ131, MQ135, MQ7, DHT11) can be installed to collect the features of air and send it to IoT Smart Objects (like Wi-Fi Modular, Raspberry Pi, Arduino Microntrolllers). These Smart Objects then transfer the data to software or cloud using the IoT Networks (like Local and Personal Area Networks (LAN/PAN), Cellular, Mesh, Low Power Wide Area Networks (LPWAN)). These data can

Attributes Sensors	Cost	Used for	Range of detection	Ref.
MQ9	₹ 100–300	CO/Combustible Gas	200~10,000 ppm	[37]
MQ131	₹1500-2500	O ₃ /Ozone	10~1000 ppm	[38]
MQ135	₹ 150–500	NH ₃ , NO _x , Alcohol, Smoke	10~300 ppm (NH ₃ /Alcohol) 10~1000 ppm (Benzene)	[39]
MQ7	₹ 170–300	CO	20~2000 ppm	[40]
DHT11	₹ 150–5000	Temperature & Humidity	0~50°C (Temperature) 20%~ 80% (Humidity)	[41]

Table 5 Names and Attributes of few Sensors

be sent to the front-end using API (Application Programming Interface) Integration. Further, it can be used for analysis and forecasting [42].



Fig. 6 Working of IoT based Models

Borges et al. in [11] used the LM317 integrated circuit (positive voltage regulator), MQ7 sensors to measure CO gas, an ESP8266, which has the advantage of transferring data wirelessly by using TCP/IP protocol. With ESP8266, ESP8266-12E is also presented in their setup, which was assisted using Arduino sketch via USB (Universal Serial Bus).

Ayele et al. in [15] have used a DHT11 sensor for real-time humidity and temperature data. The MQ135 sensor is used for generating gas data, and both sensors are connected with the ESP8266 microcontroller to send the data to the webserver. Whereas, Kiruthika et al. [43] used Raspberry Pi as the mainboard, which is interfaced with the MQ5 gas sensor and time-humidity sensor, and whole data is transmitted via ESP8266 to the webserver. The python script is used to code. If humidity, temperature, or gas value crosses the threshold values, the actuators are turned ON. If it is below the threshold value, then it is stored in the MySQL database and queried via ThingSpeak. This setup is a truly low-cost, effective air monitoring system.

In [12], Moses et al. used various sensors, MQ131 for Ozone, MQ7 for CO, 110–602 Sulphur dioxide electrochemical sensor, NO₂, and PM_{2.5} sensors. These all are connected to NB-IoT (NarrowBand-Internet of Things) and Raspberry Pi. NB-IoT collects the data for the sensor and sends it to the network via a narrow band device, and by installing MQTT protocols, the NB-IoT can directly send data to the cloud service. After this, the data is used to calculate AQI, and the estimated AQI is updated on Google Maps using an API. The authors have considered the maximum number of pollutants for monitoring the air.

Srivastava et al. [44] used HPMA115S0 sensor for $PM_{2.5}$ and PM_{10} , SHT10 for temperature, humidity, and CO, and Raspberry Pi 3 Model B as the microcontroller. Data collected is stored on the server, and as soon as any of the pollutants exceeds its threshold values as declared by Central Pollution Control Board (CPCB), it sends an emergency notification on the android application.

Okokpujie et al. [45] used an MQ135 sensor connected through Arduino to ESP 8266 Wi-Fi Module. A 16 by 2 LCD Screen is used to display the data locally. Data was sent to the ThingSpeak server, and graphs were plotted to study the air quality. In [46], Gupta et al. proposed an air monitoring system using IoT hardware, ThingSpeak, and android application. DHT11, MQ-2, and SDS021 sensors were used to collect the pollutant values (i.e., Temperature, Humidity, CO, Smoke, LPG, $PM_{2.5}$ and PM_{10}). These sensors are then con-

nected to Raspberry pi, and an Analog-Digital Converter is also connected for converting analog signals to digital ones. These data are sent and stored on ThingSpeak via API on the private channel. On ThingSpeak, the data is plotted against the date and time. This data on ThingSpeak will create three JSON files which will later be fetched by android studio by JavaScript Object Notation (JSON) parsing. Firebase API is then included in Android or iOS applications to get the real-time database, storage, analytics, and other data. Users can access pollutants concentration for any date and time.

An interesting model was also shown by Esfahan et al. [47], who proposed an IoT-based indoor air pollution monitoring system. Based on the US Environmental Protection Agency (US EPA) guideline, the indoor air quality index (IAQI) was taken into account. Sensors were connected to ESP32 based microcontroller, which transfers the data to the server using Wi-Fi. A fan is also connected to cool the setup if needed. The data collected is sent to the third-party application Blynk on smartphones. Then two experiments were carried out; the first one was carried out in computer laboratories of the School of Engineering, University of Warwick. The result with 100 occupants showed a poor level of CO_2 . The second experiment was done in a kitchen where food was being prepared with an occupant. CO_2 levels were high, whereas $PM_{2.5}$ and TVOC (Total Volatile Organic Compounds) levels were also high but were at an acceptable limit.

3.3 Optimal IoT Setup Placement Studies

The optimal placement of IoT setup means how a proper IoT network can be developed across the targeted area to get more precise results. It is an essential aspect of effective and efficient air monitoring systems. In this direction, Sun et al. [13] proposed and optimized the sensor placement strategies in Cambridge city, U.K. The objective of this study includes the Monitoring of traffic emissions, protection to vulnerable sections of citizens, and maximizing the satisfaction of citizens. Dataset of this study comprises Traffic pattern data, population data, and data where vulnerable people spend their most time. The authors suggested reorienting the present sensor placement pattern in Cambridge and concluded the study on two points: first, to increase the number of low-cost air pollution setups, and second, the sensor setups must be spread uniformly across the city to fulfill the objectives. Optimal sensor placement is a fundamental and widespread optimization-based problem. Many authors have investigated the optimal placement of sensors. For example, the studies on the Gaussian Process model for optimal sensor placement by Dur et al. [48], Krause et al. [49], Longi et al. [50], and [51].

4 Big Data and Machine Learning-Based Models

4.1 Big Data

Big data is not simply the data big in size that requires big storage; in fact, it is one of its major properties. It is a complex, unstructured, rapidly generating data huge in volume [52]. Hrehova [53] has presented some definitions of big data from various sources. Generally, big data is defined in terms of its V's concepts. Initially, it was defined with three V's concepts: Volume, Velocity, and Variety by Doug Laney [54], and then expanded to include

fourth V: Veracity [55]. At present, it includes more than twenty V's [56]. Big data imposed new challenges like the scalability and efficiency of the traditional ML and data analytics algorithms for big data. Big data analytics provides the solution by implementing these traditional algorithms on various big data platforms [57].

Fig. 7 describes the 4 V's definition of Big Data. Volume refers to data size, Variety means the kind of data generated (structured, semi-structured, unstructured), Velocity is the speed of generated data, and Veracity is the grade until data can be trusted.

4.2 Machine Learning (ML)

ML is the process of training the computer so that it can automatically improve its performance over time without being explicitly programmed for a specific task. The machine learns using various algorithms present, and then it is trained on the particular amount of data known as train data, and later its performance is evaluated on an entirely new set of data that is not encountered before, known as test data. Some applications of machine learn-



Fig. 7 4 V's Concept of Big Data

ing are pattern recognition, image analysis, predictions, recommendation, etc. [58]. ML can broadly be divided into three categories which are shown in Fig. 8.

i) Supervised Learning

In Supervised Learning, labeled data are used for training purposes. Supervised learning algorithm includes Regression and Classification algorithms. Regression algorithms are mainly used for predicting an entity and determining the relation between quantitative data. It has explicitly Linear Regression, Decision Trees, Bayesian Networks, and Fuzzy Classification. Classification algorithms are used for classifying the set of data into new classes, and it includes Logistic Regression, Classification Tree, Random Forest, and Support Vector Machines (SVM) [58], [59].

ii) Unsupervised Learning

It is just the opposite of Supervised Learning. No labeled or solved data is present in the training set. The machine has to find out its solution. It mainly includes Dimension Reduction and Clustering algorithms. Dimension Reduction algorithms reduce the large dataset with many features into data with fewer features while maintaining the fundamental aspect. Principal Component Analysis (PCA), Tensor Reduction, Random Projection, and Multidimensional Statistics are included in Dimension Reduction. Whereas Clustering algorithms partition the input data into clusters based on certain criteria. It includes Gaussian Mixture Models, Genetic Algorithms, Hierarchical Clustering, and K-means Clustering [58], [59].

iii) Reinforcement Learning

In Reinforcement learning, the agent learns to make decisions over time by consequences. It uses trial & error using feedback methods to learn. Like Supervised Learning, it also uses the mapping between input and output but uses punishment and rewards as feedback. The two important models for reinforcement learning are Markov Decision Process and Q Learning. The applications of Reinforcement learning are self-driving cars, gaming, recommendation system, financing, trading, etc. [60], [61].

4.3 Machine Learning Over Classical Statistical Forecasting Methods

Classical statistical forecasting methods are used for univariate time series problems. Some of the classical statistical methods for time series are as follows and the same is summarized in Fig. 9.





- Naïve 2: This method sets the forecast to last observation is usually helpful if the data consists of long periods of apparent ups and downs. Hence, it is also known as the adjusted random walk model for the season [62].
- ii. Simple Exponential Smoothing (SES): This method is suitable for the dataset with no clear seasonality or trends [62].
- iii. Holt: It is an extension of SES for the data with trends, and to capture the seasonality, Holt-Winter method is used [62].
- iv. **Damped Exponential Smoothing (DES):** Previous methods assume that trend would go on forever, and to eradicate this idea, Damped exponential smoothing is done [62].
- v. Theta Method: This method decomposes the original data into seasonality-based theta lines [63].
- vi. **ARIMA:** ARIMA stands for the Auto-Regressive Integrated Moving Average and aims to correlate features. It is a widely used approach for time series problems [62].
- vii. ETS: ETS stands for Error, Trend, and Seasonality. It is an exponential smoothing model in which decomposition plots help to find out whether to add, multiply, or leave out these trends, errors, and seasonality [62].
- viii. HMM: HMM stands for Hidden Markov Model, a statistical Markov model that detects the hidden states to learn about Markov Chains. It is a probabilistic model used to derive the probabilistic aspects of random processes. It's one of the important use is in the field of Natural Language Processing for Part of Speech Tagging [64], [65].

The classical models can give a linear behavior between the target and the independent feature, whereas Machine Learning algorithms-based approaches can detect the non-linear behavior of the data set without knowing any other facts about the dataset. Moreover, these ETS or ARIMA models are the local model for each time-series model. In contrast, ML-based models learn jointly over the whole series. Hence, if a multivariate dataset, Machine Learning is a good choice over the classical methods [66] [67]. In other models like forecasting shoreline evolution for sandy coasts [68] or predicting exchange rates [69], impressive results have set the shift of the time series model from classical method to these Machine Learning-based methods.

Table	e 6 Comparison of Different System	s with Big Da	tta Technologies and Machine L	earning Algorithms		
Ref.	Problem statement	Region	Parameters	Data-source	Technique	Results
[11]	IoT and Big Data techniques- based study of CO	Sao Paulo, Brazil	CO	IoT setup (Sensor: MQ7)	Apache Hadoop and MapReduce through R studio.	Sao Paulo's CO readings were double what was established by WHO.
[15]	A monitoring and predicting system for air pollution	Vadodara, India	Temperature, humidity, and gas	IoT setup (Sen- sors: DHT11 and MQ135)	LSTM algorithm for prediction	The LSTM curve showed experiment accuracy.
[12]	Air pollution monitoring system based on IoT, and re-routing system based on ML	Coim- batore, India	Ozone, CO, SO ₂ , NO ₂ , and PM _{2.5}	IoT setup (Sensors: MQ131 MQ7 110–602, NO ₂ and PM _{2.5} sensor)	Neural Network and SVR	Using this system Passenger was getting the re-routed less polluted way.
[4]	Study of ML algorithm to give improved IoT-based monitoring system	Rourkela, India	PM _{2.5} , PM ₁₀ temperature, humidity, and CO	IoT setup (Sensors: HPMA115S0, SHT10)	SVR and RFR algo- rithm models	RFR gave better results.
[71]	An efficient air quality analy- sis system using advanced technologies	Beijing, China	AQI, CO, NO ₂ , PM _{2,5} , PM ₁₀ , city, date, wind speed	Chinese online Air quality monitoring platform.	HDFS, Spark MLlib for Back Propagation neural network	Cluster-based severity and pollutant vs. year were visualized.
[30]	A deep learning air quality pre- dicting model	Ankara, Turkey	Nitrogen Dioxide	CityPulse EU FP7 Project	SVM and LSTM	With precision and F1-Score 98% and 97%, respectively, LSTM gave better results.
[72]	A better air quality predicting system using Machine learning Algorithms	India	SO, NO _X , HC, and Humidity	CPCB, Govern- ment of India	K-Means clustering to divide the data into 3 categories based on pollutant concentra- pin then model is compared using Decision tree and Multinominal logistic regression.	Multinominal logistic regression performed better with an error rate of 0.428.

Table	6 (continued)					
Ref.	Problem statement	Region	Parameters	Data-source	Technique	Results
[22]	Comparative study of four advanced regression	Shenyang, Beijing, Shanghai, Guang- zhou, and Chengdu (China)	Meteorological variables and PM2.5	Heterogenous sources were used. Sensors collected data of O ₃ , NO ₂ , SO ₂ , and PM.	GBR, DTR, MLP, and RFR	RFR performed best out of all based on processing time, MAE. and RMSE with hyper-parameter tuning.
[73]	A promising Machine Learning Model for predicting AQI of New Delhi	New Delhi, India	PM ₁₀ , PM _{2.5} , NO ₂ , SO ₂ , CO, O ₃ , NH ₃ , Pb, and Meteoro- logical variables.	CPCB, Govern- ment of India	Neural Networks and six kinds of SVMs	Results were evaluated based on accuracy: Medium Gaussian SVM got 97.3% whereas for Neural Network, 91.62%.
[31]	Air quality prediction model using IoT and big data analytics	Morocco	NO2	Kaggle	Databricks File Sys- tem; GBTs as well as ML pipelines	RMSE is equal to 0.13; hence the model is accurate.
[14]	IoT-based Monitoring and predict- ing system	China	Smog, Inhalable particles, CO, Cl, HCl, and many other environmental and meteorological factors	IoT setup	Progressive regres- sion neural networks	Meteorological inputs improved the prediction and the amount of data sent as input also affect the efficiency.
[74]	A multivariate analysis using ML model to investigate the influential factors of air quality	U. S.	Meteorological, Energy, Economical, Demographi- cal, Transportation, and Environmental	GIS	MLR, LR, CART, KNN, SVM, ANN, BB LR, BB SVM, RF, GBDT, DNN, XGBoost	XGBoost performed better than others.
[75]	ML predictive model to forecast PM _{2.5} concentration for air quality	Taiwan	Meteorological, time-series data	Air pollution stations	CART, RFR, GBR, KNR, MLPR	GBR performed better than others.
[76]	Deep Learning model for predict- ing PM ₁₀ and PM _{2.5} .	South Korea	Fine dust particles (PM_{10} and $PM_{2,5}$), Location Data, and Meteorological data.	Air Korean Web- site and Korean meteorological agency website	Stacked Auto Encoders	Performed best for Gwangju city with RMSE for PM_{10} and $PM_{2.5}$ is 5.16 and 2.18 respectively.
[12]	Seasonality-based PM level prediction	Ghazi- abad, India	11 attributes including Meteorological and pollutant data	CPCB, Govern- ment of India	GBM, MLP, SVR- RBF, RF	GBM outperformed all and CO concen- tration affected the prediction of GBM the most.

Table	6 (continued)					
Ref.	Problem statement	Region	Parameters	Data-source	Technique	Results
[78]	Improved PM _{2.5} predicting model	Beijing City, China	PM _{2.5,} Temperature, Dew point, WS, WD, and Timing data.	UCI Machine Learning Reposi- tory US	BiLSTM	The model performed better than exist- ing models. The evaluation was done using RMSE, MAE, and SMAPE.
[62]	$PM_{2.5}$ and PM_{10} forecasting model for Bogotá	Bogotá, Colombia	Wind Speed, Wind Direction Temperature, Precipitation, Relative Humidity, And Solar Radiation.	RMCAB	PCA, K Means, BPNN	The model was evaluated using RMSE, MAE, and CC.
[80]	A new model based on the LSTM method for predicting PM _{2.5}	Taiwan	17 attributes including local data, near station data, and industrial data.	TEPA and CWB	ALSTM	The model improved the accuracy with RMSE aspect 1.59, 0.44, and 2.77 and with MAE aspect 1.16, 0.91, and 5.27, better than existing models.
[81]	A hybrid NARX model for pre- dicting PM _{2.5}	Beijing, China	PM _{2.5} , cumulated wind speed and hours of rain	UCI Machine Learning Reposi- tory US	Hybrid NARX	The model was evaluated based on many metrics but NARX-LSTM and NARX- XGBRF were best.
[83]	Algorithm to improve multiclass imbalance in dataset	Delhi, India	PM ₁₀ , SO ₂ , PM ₂₅ , O ₃ , NOX, NO ₂ , NO, NH ₃ , CO, AQI, WD, C ₆ H ₆ , WS, RH, SR, BP, and AT	CPCB, Govern- ment of India	Scalable kernel-based SVM	The proposed model performed best with 99.66% accuracy.
[85]	AirRL: a reinforcement learning model for inference of urban air quality	Beijing, China	$PM_{2.5}$ and PM_{10}	36 Monitoring Stations	Reinforcement Learn- ing and DNN	AirRL performed best with accuracies of 71.37% and 64.99% for PM _{2.5} and PM ₁₀ respectively.



Fig. 10 Distribution of different algorithms used by various researchers



Fig. 11 Countries whose datasets are considered for air pollution models under study

4.4 Big Data and Machine Learning-Based Monitoring and Forecasting Systems

The data obtained from the real-time mechanism or downloaded needs to undergo a preprocessing architecture to get improved and reliable results. The data obtained from sources have some noise, missing values/attributes, or have errors. The process of cleaning the raw data is known as Data Pre-Processing or Data cleaning. After data is cleaned, data from all the sources is combined and stored; this huge data is managed by big data tools. After extracting all good quality data, target data mining is performed using machine learning algorithms.

In [11], Borges et al. collected the CO data for Sao Paulo's Metropolitan Area (SMA) by four IoT setups, and data was directed to Apache Hadoop through R studio. Using a popular HDFS (Hadoop Distributed File System) tool, MapReduce was implemented to process and analyze the data. Using Shiny, an R package that builds an interactive web application visualizes data. The data from all the sensor setups were compared, mean values per day for every sensor, sensor density, and summary of the measurements were potted using Shiny. It was found that the SMA's CO value was double the value established by WHO.

Ayele et al. [15] proposed real-time Monitoring and predicting system. Data was collected from the IoT setup, the collected data was stored on the web server, and the Long Short-Term Memory (LSTM) algorithm was used for prediction [70]. This experiment was performed by python 3.6.3 with TensorFlow as a backend. This experiment was concluded with good accuracy.

Moses et al. [12] collected data by IoT setup and suggested users with the alternative less polluted route on Google Map. Time series prediction of air quality was done by Neural Networks (NN) and Support Vector Machine (SVM) regression algorithm. For predicting AQI value, NN was used with sigmoid activation function, whereas SVM helps in error tolerance by individualizing hyperplanes.

Srivastava et al. [44] collected data via IoT setup and considered Central Pollution Control Board (CPCB) threshold values to send an emergency notification on the android application. Predictive analysis is done by two algorithms, Support Vector Regression (SVR) and Random Forest Regression (RFR). The models respective to these algorithms were conducted and evaluated using precision, recall, support, and F1 score. The accuracy in the SVR model was 90%, whereas in RFR it was around 99%. RFR performed well because many individual decision trees are involved in the algorithm, improving the overall result.

Wang et al. [71] proposed a data analysis and forecasting system for air pollution. The data used was taken from the Chinese air quality online monitoring and analysis platform using an API key. For storing atmospheric data, HDFS was used, while for calculation engine for industrial data, and Spark was used. Data were pre-processed to remove the duplicated or erroneous data. Data mining was done using Backward Propagation (BP) neural network. BP is a supervised learning algorithm, popularly used for prediction so that weights can be repeatedly adjusted and output is much closer to the expected vector. Data obtained through python crawler – AQI, CO, NO₂, PM_{2.5}, PM₁₀, city, date, wind speed, etc. was input to BP neural network, and it was found that wind speed value is one of the critical factors. The results were displayed using a visualization platform. This system used a combination of various technologies and proposed out an efficient method.

Kök et al. [30] have given a promising deep learning model for smart cities. Data used were collected from the CityPulse EU FP7 Project of Aarhus and Brasov cities in Denmark and Romania, respectively. Ozone and NO₂ pollutants were considered for this project. The dataset was divided into 69.5% and 30.5%, and python platforms Keras DL framework and TensorFlow were used. The data was trained on two algorithms, SVM and LSTM, for prediction. LSTM is a popular Recurrent Neural Network (RNN) that has feedback connections. Both models are compared by Root Mean Squared Error (RMSE) and Mean Absolute Error (MAE), followed by Confusion matrix for each model, precision, recall, and F1 score. The model with nearly good predictions for red alarms is considered yellow and green. Best

precision and F1-Score as 98% and 97% respectively were given by LSTM, whereas for SVM, values were 95% (Precision) and 96% (F1 score) for red alarm. This system gave efficient and promising results for IoT data collected through CityPulse EU FP7 Project.

Nandini and Fathima [72] collected pollutant data and meteorological data from Central Pollution Control Boards (CPCB) and the Meteorological department. Then dataset was divided into validation data and test data as 90% and 10%, respectively. Then using K-Means Clustering, the data was converted into 3 clusters- Good, Moderate, and Unhealthy. After that, Multinominal Logistic Regression was used to observe the pattern between factors and results, and a decision tree was also used as a support tool to improve the conditional control statements. A confusion matrix is used to analyze the performance of both algorithms. The error rate value for the Regression model was found to be 0.428, and that of the Decision tree model was 0.666. Hence, the regression model was the best fit model.

In [22], Ameer et al. presented a comparative study for four advanced regression algorithms. The dataset consists of five cities of China – Shenyang, Beijing, Shanghai, Guangzhou, and Chengdu for a period from January 01, 2010, to December 31, 2015. Features of the dataset include $PM_{2.5}$ readings and other meteorological data of specific cities. By observing data, $PM_{2.5}$ had a negative correlation with other features, so data of every city was converted into correlation matrices. Prediction analysis was done using four advanced regression algorithms – GBR (Gradient Boosting Regression), DTR (Decision Tree Regression), MLP (Multilayer Perceptron Regression), and RFR (Random Forest Regression), and the results were evaluated by MAE, RMSE, and processing time. Decision Tree Regression is simple and took less processing time with MAE and RMSE, which is between 8 and 21% and 0.06 to 0.24, respectively. MAE for Random Forest Regression ranged between 6 and 18%, whereas RMSE ranged between 0.05 and 0.18, and processing time was lower than GBR and MLP. GBR had the highest error values. MLP was comparable to RFR. Out of all, RFR performed well after hyperparameter tuning using Spark. Hyperparameter tuning was done using a 10-fold cross-validation method and GridSearchCV function.

Mahalingam et al. [73] also studied the machine learning algorithms and their performance. Dataset was downloaded from Central Pollution Control Boards (CPCB) for New Delhi, India. Then whole data was divided into six categories – Good, Satisfactory, Moderate, Poor, Very Poor, and Severe. Neural Networks do prediction analysis with eight neurons in the hidden layer and six types of Support Vector Machines (SVM). Six SVMs with their accuracies are – Linear SVM (89.2%), Quadratic SVM (94.6%), Cubic SVM (94.6%), Fine Gaussian (62.2%), Medium Gaussian (97.3%) and Coarse Gaussian (78.4%). Neural networks provide an accuracy of 91.62%. Hence, Medium Gaussian is found the best fit.

In [31], Alaoui et al. proposed a model of air pollution using big data and ML algorithms. The data used is consists of NO₂ attributes (NO₂ units, NO₂ mean, NO₂ AQI, NO₂ 1st max. value, and NO₂ 1st max. hour) and meteorological data, which was taken from Kaggle. This data was pre-processed and was stored on Databricks. Then it was loaded and split into training and testing sets as 70% and 30%, respectively. Gradient-boosted trees (GBTs) as well as ML pipelines were used for data modeling and evaluated by Root Mean Squared Error (RMSE). Databricks were used for cloud-based data handling purposes. The obtained value of RMSE was 0.13. Hence the model is considered accurate.

In [14], Xiaojun et al. presented an IoT-based system. Environmental, as well as meteorological features, were taken into account. Two IoT setups were installed for other months – one was for January, February, November, December, and another for April, May, June, July, August, September, October. The ratio of Training Set: Validation Set: Test Set taken was 2:1:1. Data mining was done through neural networks (Input layer -24, Hidden layer -4, and Output layer -1). Firstly, five meteorological factors were included, and 90% confidence was achieved using progressive regression) latter on input nodes were increased to 29 whereas hidden layer nodes to 6, and a study was done to compare the model's performances with and without meteorological factors. Another study was done by establishing artificial neural networks, and data of 5 years was taken as input. The model was studied based on the amount of data carried as input. Three years of data gave more close results than five years of data. It was concluded that meteorological data and the amount of data chosen as input data affects the results.

In [74], Ma et al. analyzed 171 features to investigate the factors that influence air quality. They have used non-linear ML algorithms MLR (Multiple Linear Regression), LR (Logistic Regression), Decision Tree-CART, kNN (K Nearest Neighbors), SVM (Support Vector Machine), ANN (Artificial Neural Network), BB (Bagging and Boosting), LR, BB SVM, RF (Random Forest), GBDT (Gradient Boosted Decision Trees), DNN (Deep Neural Network), and XGBoost (Extreme Gradient Boosting) to find the relationship between variables. In [75], Doreswamy et al. analyzed five different ML techniques, Decision Tree-CART, RFR (Random Forest Regressor), GBR (Gradient Boosting Regressor), KNR (K Neighbors Regressor), MLPR (Multilayer Perception Regression), to forecast the $PM_{2,5}$. They have used the TAQMN (Taiwan Air Quality Monitoring Network) dataset containing the data on air pollution in Taiwan from 2012 to 2017. The data are collected from 76 stations that are located at different places in the cities. The data is further pre-processed to fill the missing values. Further, metrological and air pollution data is fed to ML techniques to forecast the $PM_{2.5}$ level. The performance of the model is compared by using crossvalidation and statistical metrics. They have found that GBR has outperformed as compared to other ML algorithms.

Xayasouk and Lee [76] developed an interesting model for PM_{10} and $PM_{2.5}$ prediction. They took a dataset from January 01, 2017, to December 31, 2017, with 12 attributes – Wind Speed, Wind Direction, Temperature, Humidity, Rain, PM_{10} , $PM_{2.5}$, and Location information of cities Busan, Daegu, Daejeon, Gwangju, Incheon, Sejong, Seoul, and Ulsan. The model used for training and prediction purposes was the Stacked AutoEndorers (SAEs) Model. AutoEndorers are a type of neural network that reproduces its input and have one of each input, hidden, and output layer. If they are stacked in a hierarchical manner, they are known as SAEs. Data is trained in a greedy layer-wise manner. For each city, PM_{10} and $PM_{2.5}$ were calculated, and RMSE was taken as performance metrics. Gwangju city performs best with 5.16 RMSE for PM_{10} and 2.18 for $PM_{2.5}$. Results were even shown by plotting the actual and predicted values.

Barthwal et al. [77] presented an interesting solution. They worked on the data of Vasundhara Station, Ghaziabad, India, which was collected from the CPCB website. Data was consisted of both meteorological and air pollutant concentrations and taken at the hourly and daily average. The duration of data was from January 17, 2019 – January 30, 2020, which covers all seasons of that region. Meteorological data includes the attributes - Atmospheric Temperature (AT), Relative Humidity (RH), Solar Radiation (SR), Wind Speed (WS), Wind Direction (WD), and Atmospheric Pressure (AP) whereas air pollutant concentrations of CO, $O_{3,}$ and SO₂. The missing values from the dataset were taken care of through Linear interpolation. The main issue discovered was the inconsistency in the correlation

between $PM_{2.5}$ and PM_{10} . There was a strong correlation in the season of Monsoon – Winter, whereas, for Spring – Summer, it was poor. So, the forecasting models used were Multiple Linear Regression (MLR), Random Forests (RF), Support Vector Regression - Radial Basis Function (SVR-RBF), and Gradient Boosting Machine (GBM) due to non-linearity in the trend. Moreover, to improve the model performance and robustness, Hyperparameters were tuned for each algorithm. The ratio of Training Set and Testing Set was 80:20, and Stratified Random Sampling was used as cross-validation. The models were evaluated based on the – Metrics (coefficient of determination (R²), Mean Error (ME), Absolute Mean Error (AME), Relative Error (RE) and Root Mean Square Error (RMSE)), Variation Importance Ranking (VIR) and Partial Plots (PPs). PPs for whole data and metrics for 7-days prediction for each season revealed GBM as the best model out of all. VIR analysis on the GBM algorithm showed CO as the most important feature for predicting PM levels.

Jeya et al. [78] uses Bi-directional LSTM (BiLSTM) model for $PM_{2.5}$ prediction. Data was extracted from UCI Machine Learning Repository, US for Beijing City and contains hourly data from January 01, 2010, to December 31, 2014, with features – Dew point, Temperature, Wind Direction, Wind Speed, Year, Month, Day, Hour, and $PM_{2.5}$. All data were normalized and then split into training and testing datasets in ratios 80 and 20, respectively. For correlation, a heatmap was plotted, and features with small correlations were removed for better accuracy. For training, BiLSTM was used with 32 neurons, batch size-16, epochs-20, Adam optimizer, and drop out is 0.2. The loss curve of the training set was plotted for the modeling of the test set accordingly. The Evaluation was done using RMSE, MAE, and Symmetric Mean Absolute Percentage Error (SMAPE). The RMSE of the proposed model was 9.86, whereas MAE and SMAPE were 7.53 and 0.1664, respectively, which was better than previously existing models.

Franceschi et al. [79] presented a prediction model of PM2.5 and PM10 based on Artificial Neural Networks (ANN) and k-means clustering for the city of Bogotá (Colombia). The data was collected from the Red de Monitoreo de Calidad del Aire de Bogotá (RMCAB) website for five years and from 13 stations. Missing data of the dataset can be considered up to the limit of 15% due to the device calibration, rain, or any other factor that could have caused the error or missing values in the collected data, and then data was normalized on the scale of [0, 1]. Principal Component Analysis (PCA) was used to reduce the dataset into less dimensional data while maintaining the main characteristics of the original dataset. Criteria of relevance were chosen as Cattell's Scree Test and featured with an eigenvalue of more than 0.2. The Back Propagation Neural Network (BPNN) model is considered as a forecasting model, whereas K means clustering groups the data before sending it to the forecasting model. X means an algorithm that is faster and starts with an assumption of the minimum number of clusters was chosen to find out the best number of clusters for each station. Quantitative analysis was done for measuring the performance, RMSE, MAE, and correlation coefficient (CC) values were 10.56, 14.01, and 0.72 respectively for PM₁₀, while for PM_{2.5}, it was 7.19, 9.34, and 0.66 respectively.

Chang et al. [80] recommended the system for $PM_{2.5}$ forecasting for Beijing, China. Data were collected from Taiwan Environmental Protection Agency (TEPA) and Central Weather Bureau (CWB) with 17 attributes for five years (Training Data) and 1-year data (Testing Data). Data pre-processing is done using Akima Normalization. The Aggregated LSTM (ALSTM) is used as a forecasting model with three input datasets (The local dataset, the neighboring station dataset, and industrial stations datasets) passed through two alternating

LSTM and Dropout and finally to Dense, Merge, Dense, Output layer. Dropout layers are added to avoid overfitting of the model. This model was compared to LSTM, GBT, and SVR based on MSE and RMSE of 8 sequential future hours. ALSTM outclass other algorithms with minimum errors. Later the performance was also measured on a region-wise basis using MAPE, RMSE, and MAE.

Moursi et al. [81] proposed the Hybrid Non-linear AutoRegression with eXogenous input (NARX) model [82], which is mainly used for time series modeling. Dataset was taken from the University of California, Irvine website, which published the data from Beijing, China, for five years (2010–2014). The training segment took the first four years of the dataset, whereas the testing segment took the last year of dataset. Features used for this model were PM_{2.5} value, cumulated wind speed and hours of rain. Hybrid NARX model was run on both a regular PC as well as on Raspberry Pi 4 but each time the model can take one ML model. The ML model taken into consideration were Long Short-Term Memory (LSTM), Random forests (RF), Extra Trees (ET), Gradient Boost (GB), Extreme Gradient Boost (XGB) and Random Forests in XGBoost (XGBRF). The evaluation of performance was done using Root Mean Square Error (RMSE), Coefficient of Determination (R²), Index of Agreement (IA) and Normalized Root Mean Square Error (NRMSE). The NARX- LSTM model worded out beat in terms of performance. But for accuracy and efficiency, NARX-XGBRF turned out best among all others.

Ketu et al. [83] propounded an algorithm to manage multiclass imbalance in the dataset using the Adjusting Kernel Scaling method (AKS), which is integrated with SVM classification. Dataset was taken from CPCB for Delhi region. In the algorithm, the kernel transformation function is calculated. This is done by calculating every support vector's weighting factor and parameter function at every iteration using Chi-square test [84]. The model was compared with already existing classification models, ADB (Ada Boost Algorithm), MLP (Multilayer Perceptron Algorithm), GNB (Gaussian NB Algorithm), Standard SVM (Support Vector Machine Algorithm) but out of all, the proposed model gave out best result with 99.66 accuracy. To avoid overfitting, K-Fold validation was used.

Zhong et al. [85] proposed a reinforcement learning based model named AirRL for the analysis of urban air quality. The authors claimed that their model is the first one to apply reinforcement learning for air pollution inference. The model has two modules: a station selector module that dynamically selects monitoring stations and a regressor module for air quality inference. Station selection is formulated using reinforcement learning to select optimal stations. The regressor module uses DNN to learn the relations among complex features. The authors recorded the data at 36 monitoring stations every hour during the period from 01/05/2014 to 30/04/2015 in Beijing, China. They used the dataset of only 30 stations which have fewer missing values for the analysis of $PM_{2.5}$ and PM_{10} . AirRL performed best with accuracies of 71.37% and 64.99% for $PM_{2.5}$ and PM_{10} respectively against the other baseline methods considered for comparison. The RMSE values of AirRL were 35.7555 and 40.8122 for $PM_{2.5}$ and PM_{10} respectively, which was better than the other baseline methods.

Table 6 represents the various ML and Big Data models summarized, whereas Fig. 10 shows the distribution of multiple algorithms that gave better results to researchers. SVR is giving more stability to the predicting model than any other algorithm. NNs and LSTMs are emerging as a better solution for the problem because weights are assigned to each input feature, and nodes are set with the thresholds, which gives a better result than any other algorithm.

Further, Fig. 11 shows the countries whose datasets have been taken into account for predicting purposes. Table 7 distinguishes the above-discussed ML models in various aspects. i.e., is data collected with self-installed sensor setups, how missing data is handled, does the correlations were taken into account, and finally, hyperparameters were tuned or not.

5 Research Issues and Challenges

After examining and reviewing the literature in the above sections, some research issues and challenges for designing and implementing efficient models for air pollution monitoring and forecasting are underlined. Forecasting is a technique that is used consciously or unconsciously to predict what will happen and what is the likelihood of specific events. The research issues and challenges discussed below are also potential directions for future research.

- (i) **Quality of Data:** The IoT infrastructure collects the data, but sometimes, due to poor network, sensor qualities, and connection faults, the data quality degrades. Good data quality with fewer missing and erroneous entries will give out the results with better accuracy.
- (ii) Quantity of Data: In [73], Mahalingam et al. considered data records for one month (December 1 to 31) whereas Xiaojun et al. [14] highlight the amount of data considered is also an important factor for the accuracy of the model. The models must be tested on varying quantities of data. An appropriate amount of data must be considered for a better model.

Table 7 Comparing of models on various parameters	Ref.	Sensor setup	Handling miss- ing data	Correla- tion between parameters	Hyperpa- rameter tuning
	[12]	\checkmark	-	_	\checkmark
	[13]	\checkmark	\checkmark	\checkmark	-
	[15]	\checkmark	-	-	-
	[22]	\checkmark	\checkmark	\checkmark	\checkmark
	[44]	\checkmark	\checkmark	-	\checkmark
	[71]	-	\checkmark	-	\checkmark
	[30]	-	-	-	\checkmark
	[51]	-	\checkmark	-	-
	[73]	-	-	\checkmark	-
	[53]	-	\checkmark	-	-
	[74]	-	\checkmark	\checkmark	\checkmark
	[75]	-	\checkmark	-	-
	[<mark>76</mark>]	-	-	\checkmark	-
	[57]	-	\checkmark	\checkmark	\checkmark
	[58]	-	\checkmark	\checkmark	-
	[<mark>79</mark>]	-	\checkmark	\checkmark	\checkmark
	[<mark>80</mark>]	-	\checkmark	\checkmark	-
	[81]	\checkmark	-	-	-
	[83]	-	-	\checkmark	\checkmark
	[85]	\checkmark	-	\checkmark	-

- (iii) Real-Time Integrated Model: A whole real-time integrated air quality monitoring and predicting system based on big data technologies and machine learning still lacks in the present scenario. Various factors affect the air quality of a particular area. To tackle all the dynamic changes, a stable and integrated model is still needed.
- (iv) Meteorological Factors: Xiaojun et al. [14] also highlighted the importance of meteorological factors in such models. The accuracy of models could be improved by considering meteorological data of that specific area with pollutant data.
- (v) Uniformity of Sensor Setups: Non-Uniform distribution of sensors across the cities affects the data quality. Sun et al. [13] underlined the importance of uniformity of sensor setups across the city for Monitoring and analyzing the city's air quality. To measure the air quality of a city, data collecting sensor setups must be kept at uniform distances.
- (vi) Number of Sensors Setups: For collecting the air quality data, a good number of IoT setups must be installed across the region [13]. This will improve the overall stability of the model as well as data quality for mining purposes.
- (vii) Processing Time of Models: The processing time of Machine learning models is also an important factor. In few proposed systems, machine learning models were solely evaluated on the accuracies and errors. An algorithm with high accuracy rates and a low processing time is preferred for an efficient air pollution forecasting and monitoring model.
- (viii) Number of Pollutants Taken into Account: The air quality of the regions depends on the concentration of various toxic gases and particles. Some papers considered few gases as input for evaluating AQI. Ideally, all the gases causing air pollution should be taken into account. An efficient model is still required to consider all the intoxicants in the air.
 - (ix) *Checking Correlations*: ML models can be improved if the correlation between features can be determined before modeling it. The features with poor correlation with the target feature can be removed, resulting in improved performance of the model.
 - (x) *Hyperparameter Tuning*: Each data has its particular behavior; adjusting hyperparameters could help in optimizing the performance of the ML model.

6 Conclusion and Future Work

Overexploitation of nature is reverting to earth's creatures in the form of deadly outcomes. The quality of air needs constant attention and evaluation. The advanced communication, computation, and analytics technologies, i.e., Internet of Things infrastructures, Big Data technologies, and Machine Learning algorithms, could provide us with more stable and efficient models for Monitoring and forecasting air pollution. This paper has reviewed and examined the recent air pollution monitoring and forecasting systems approach. Also, the existing model and systems have been compared on various parameters, and some significant research issues and challenges have been discussed in this paper. Further, some practical methods are suggested for improving the models. Presently, the air pollution sensing and monitoring tools and techniques suffer low efficiency, small range, and less accuracy. The real-time Monitoring and deployment of the available models are to be carried out as the future work of this study. The data-driven models can also be developed to predict, recommend, and monitor future work so that illnesses and climate change can be controlled. Finally, prevention and education can play a major role in controlling air pollution. Although, prevention and education will not reverse the adverse effects but will allow for sustainability for the future.

Acronym	Full Name
ADB	Ada Boost Algorithm
ANN	Artificial Neural Network
AQI	Air Quality Index
ARIMA	Auto-Regressive Integrated Moving Average
HMM	Hidden Markov Model
BB	Bagging and Boosting
BiLSTM	Bi-directional LSTM
BMR	Bangkok Metropolitan Region
BP	Backward Propagation
BPNN	Back Propagation Neural Network
СО	Carbon Monoxide
COPD	Chronic Obstructive Pulmonary Disease
CPCB	Central Pollution Control Board
DES	Damped Exponential Smoothing
DNN	Deep Neural Network
DTR	Decision Tree Regression
EAQI	European Air Quality Index
EPA	Environmental Protection Agency
ETS	Error, Trend, and Seasonality
EU	European Union
GBDT	Gradient Boosted Decision Trees
GBM	Gradient Boosting Machine
GBR	Gradient Boosting Regression
GBR	Gradient Boosting Regressor
GBT	Gradient-Boosted Tree
GNB	Gaussian NB Algorithm
HAQ	Household Air Quality
HDFS	Hadoop Distributed File System
IAQI	Indoor Air Quality Index
IoT	Internet of Things
kNN	K Nearest Neighbors
KNR	K Neighbors Regressor
LR	Logistic Regression
LSTM	Long Short-Term Memory
MAE	Mean Absolute Error
ME	Mean Error
ML	Machine Learning
MLP	Multilayer Perceptron Algorithm
MLP	Multilayer Perceptron Regression
MLPR	Multilayer Perception Regression

Appendix: List of Acronyms

Acronym	Full Name
MLR	Multiple Linear Regression
NAMP	National Air Monitoring Programs
NAQI	National Air Quality Index
NARX	Non-linear AutoRegression with eXogenous
NB-IoT	NarrowBand-Internet of Things
NO ₂	Nitrogen Dioxide ()
NRMSE	Normalized Root Mean Square Error
O ₃	Ozone
PCA	Principal Component Analysis
PM	Particulate Matter or Particle Pollution
PPs	Partial Plots
RE	Relative Error
RF	Random Forest
RFR	Random Forest Regression
RMSE	Root Mean Squared Error
RNN	Recurrent Neural Network
SES	Simple Exponential Smoothing
SMAPE	Symmetric Mean Absolute Percentage Error
SO ₂	Sulphur Dioxide
SVM	Support Vector Machines
SVR	Support Vector Regression
SVR-RBF	Support Vector Regression - Radial Basis Function
TAQMN	Taiwan Air Quality Monitoring Network
TEPA	Taiwan Environmental Protection Agency
TVOC	Total Volatile Organic Compounds
US EPA	US Environmental Protection Agency
VIR	Variation Importance Ranking
WHO	World Health Organisation
XGB	Extreme Gradient Boost
XGBoost	Extreme Gradient Boosting
XGBRF	Random Forests in XGBoost

Authors Contribution Conceptualization: Amisha Gangwar, Sudhakar Sing, Richa Mishra, Shiv Prakash; Methodology: Amisha Gangwar, Sudhakar Sing, Richa Mishra, Shiv Prakash; Formal analysis and investigation: Amisha Gangwar, Sudhakar Singh, Richa Mishra, Shiv Prakash; Writing - original draft preparation: Amisha Gangwar, Sudhakar Singh; Writing - review and editing: Sudhakar Singh, Richa Mishra, Shiv Prakash; Resources: Amisha Gangwar, Sudhakar Singh, Richa Mishra, Shiv Prakash; Supervision: Sudhakar Singh, Richa Mishra, Shiv Prakash.

Funding No funding was received for conducting this study.

Data Availability Not applicable.

Code Availability Not applicable.

Declarations

Conflict of interest The authors declare that they have no conflict of interest in this paper.

References

- WHO (2014). Household air pollution and health, WHO Media centre, www.who.int/mediacentre/factsheets/fs292/en (accessed Apr. 01, 2021).
- Arora, P., Rehman, I. H., Suresh, R., Sharma, A., Sharma, D., & Sharma, A. (Nov. 2020). Assessing the role of advanced cooking technologies to mitigate household air pollution in rural areas of Solan, Himachal Pradesh, India. *Environ Technol Innov*, 20, 101084. https://doi.org/10.1016/j.eti.2020.101084.
- Smog | National Geographic Society (2021). https://www.nationalgeographic.org/encyclopedia/smog/ (accessed Apr. 27 2021)
- Ha Chi, N. N., & Kim Oanh, N. T. (Feb. 2021). Photochemical smog modeling of PM2.5 for assessment of associated health impacts in crowded urban area of Southeast Asia. *Environ Technol Innov*, 21, 101241. https://doi.org/10.1016/j.eti.2020.101241.
- Iaccarino, L. (Feb. 2021). Association between Ambient Air Pollution and Amyloid Positron Emission Tomography Positivity in Older Adults with Cognitive Impairment, *JAMA Neurol*, 78(2), 197–207. https://doi.org/10.1001/jamaneurol.2020.3962.
- Gilliland, G. L. (May 1998). Human carboxyhemoglobin at 2.2 Å resolution: structure and solvent comparisons of R-state, r2-state and T-state hemoglobins. *Acta Crystallogr Sect D Biol Crystallogr*, 54(3), 355–366. https://doi.org/10.1107/s0907444997012250.
- NationalGeographic (2014). Acid rain facts. National Geographic. http://environment.nationalgeographic.com/environment/global-warming/acid-rain-overview/ (accessed Mar. 10, 2021).
- Ren, J. (May 2021). Effects of O3 pollution near formation on crop yield and economic loss. *Environ Technol Innov*, 22, 101446. https://doi.org/10.1016/j.eti.2021.101446.
- Liu, D. (May 2021). Value evaluation system of ecological environment damage compensation caused by air pollution. *Environ Technol Innov*, 22, 101473. https://doi.org/10.1016/j.eti.2021.101473.
- Gulia, S., Shiva Nagendra, S. M., Khare, M., & Khanna, I. (2015). Urban air quality management-A review, *Atmos. Pollut. Res.6*(2), 286–304, Mar. doi: https://doi.org/10.5094/APR.2015.033.
- Borges, M. A., Melo, G. F., De Massaki, C., Igarashi, O., Lopes, P. B., & Silva, L. A. (2017). An architecture for the internet of things and the use of big data techniques in the analysis of carbon monoxide, in *Proceedings 2017 IEEE International Conference on Information Reuse and Integration, IRI* Nov. 2017, vol. 2017-Jan., pp. 184–191. https://doi.org/10.1109/IRI.2017.76.
- Moses, L., Tamilselvan, R., & Karthikeyan (2020). IoT enabled Environmental Air Pollution Monitoring and rerouting system using machine learning algorithms. *IOP Conf Ser Mater Sci Eng*, 955(1). https://doi.org/10.1088/1757-899X/955/1/012005.
- Sun, C., Li, V. O. K., Lam, J. C. K., & Leslie, I. (2019). Optimal Citizen-Centric Sensor Placement for Air Quality Monitoring: a Case Study of City of Cambridge, the United Kingdom. *IEEE Access: Practical Innovations, Open Solutions*, 7, 47390–47400. https://doi.org/10.1109/ACCESS.2019.2909111.
- Xiaojun, C., Xianpeng, L., & Peng, X. (2015). IOT-based air pollution monitoring and forecasting system, in *International Conference on Computer and Computational Sciences, ICCCS 2015*, Dec. 2015, pp. 257–260. https://doi.org/10.1109/ICCACS.2015.7361361.
- Ayele, T. W., & Mehta, R. (2018). Air pollution monitoring and prediction using IoT, in *Proceedings of* the International Conference on Inventive Communication and Computational Technologies, ICICCT Sep. 2018, pp. 1741–1745. https://doi.org/10.1109/ICICCT.2018.8473272.
- Ben Atitallah, S., Driss, M., Boulila, W., & Ben Ghezala, H. (Nov. 2020). Leveraging Deep Learning and IoT big data analytics to support the smart cities development: review and future directions. *Comput Sci Rev*, 38, 100303. https://doi.org/10.1016/J.COSREV.2020.100303.
- Hajjaji, Y., Boulila, W., Farah, I. R., Romdhani, I., & Hussain, A. (Feb. 2021). Big data and IoT-based applications in smart environments: a systematic review. *Comput Sci Rev*, 39, 100318. https://doi. org/10.1016/J.COSREV.2020.100318.
- Nations, U. (2015). 70/1. Transforming our world: the 2030 Agenda for Sustainable Development Transforming our world: the 2030 Agenda for Sustainable Development Preamble, Accessed: May 06, 2021. [Online]. Available: https://www.un.org/en/development/desa/population/migration/generalassembly/docs/globalcompact/A_RES_70_1_E.pdf.
- Rafaj, P., et al. (Nov. 2018). Outlook for clean air in the context of sustainable development goals. *Glob Environ Chang*, 53, 1–11. https://doi.org/10.1016/j.gloenvcha.2018.08.008.
- EPA, O. U. S. (2019). NAAQS Table | US EPA, Us Epa, https://www.epa.gov/criteria-air-pollutants/ naaqs-table (accessed Mar. 06, 2021).
- Mintz, D. (2016). Technical Assistance Document for the Reporting of Daily Air Quality the Air Quality Index (AQI), Accessed: Apr. 30, 2021. [Online]. Available: https://www.airnow.gov/sites/default/files/2018-05/aqi-technical-assistance-document-may2016.pdf.

- Ameer, S., et al. (2019). Comparative analysis of machine learning techniques for Predicting Air Quality in Smart Cities. *IEEE Access: Practical Innovations, Open Solutions*, 7, 128325–128338. https://doi.org/10.1109/ACCESS.2019.2925082.
- Air Quality Index (AQI) Basics (accessed Apr. 30, 2021). https://web.archive.org/web/20180618144741/ https://airnow.gov/index.cfm?action=aqibasics.aqi
- Standards Air Quality Environment European Commission (accessed Apr. 30, 2021). https:// ec.europa.eu/environment/air/quality/standards.htm
- 25. van den Elshout, S., Bartelds, H., Heich, H., & Léger, K. (2012) EUROPEAN UNION European Regional Development Fund Regional Initiative Project Common Information to European Air CAQI Air quality index Comparing Urban Air Quality across., Borders-2012 Dissemination level External Component 5. Accessed: May 07, 2021. [Online]. Available: https://www.airqualitynow.eu/download/ CITEAIR-Comparing Urban Air Quality across Borders.pdf.
- The most polluted cities with the worst air quality in the world, ranked, CBS News. https://www. cbsnews.com/pictures/the-most-polluted-cities-in-the-world-ranked/ (accessed Apr. 26, 2021).
- Prime Minister's Office (2017). Swachh Bharat Abhiyan | Prime Minister of India, Government of India, http://www.pmindia.gov.in/en/major_initiatives/swachh-bharat-abhiyan/ (accessed Apr. 26, 2021).
- CPCB | Central Pollution Control Board (2021). https://cpcb.nic.in/about-namp/?&page_id=aboutnamp (accessed Apr. 30, 2021)
- National Ambient Air Quality Standards, India (accessed Apr. 30, 2021). https://cpcb.nic.in/uploads/ National_Ambient_Air_Quality_Standards.pdf
- Kök, I., Şimşek, M. U., & Özdemir, S., A deep learning model for air quality prediction in smart cities, in *Proceedings – 2017 IEEE International Conference on Big Data, Big Data 2017*, Jul. 2017, vol. 2018-Janua, pp. 1983–1990. https://doi.org/10.1109/BigData.2017.8258144.
- Alaoui, S. S., Aksasse, B., & Farhaoui, Y. (2019). Air pollution prediction through internet of things technology and big data analytics. *Int J Comput Intell Stud*, 8(3), 177. https://doi.org/10.1504/ ijcistudies.2019.102525.
- 32. Williams, R. et al. (2018). Peer Review and Supporting Literature Review of Air Sensor Technology Performance Targets. EPA
- Lazrak, N., Ouarzazi, J., Zahir, J., & Mousannif, H. (2020). Enabling distributed intelligence in Internet of Things: an air quality monitoring use case, *Pers. Ubiquitous Comput*, 1–11, https://doi.org/10.1007/ S00779-020-01483-3.
- What Is the Internet of Things (accessed Apr. 26, 2021). (IoT)? | Oracle India. https://www.oracle.com/ in/internet-of-things/what-is-iot/
- Kalajdjieski, J., Korunoski, M., Stojkoska, B. R., & Trivodaliev, K. (2020). Smart City Air Pollution Monitoring and Prediction: A Case Study of Skopje, *Communications in Computer and Information Science*, 1316, 15–27, https://doi.org/10.1007/978-3-030-62098-1_2.
- 36. Gerboles, M., Spinelle, L., & Borowiak, A. (2017). *Measuring air pollution with low-cost sensors*, European Commission
- Technical Data MQ-9 Gas Sensor (2022). https://www.electronicoscaldas.com/datasheet/MQ-9_Hanwei.pdf (accessed Nov. 12, 2022)
- MQ131 Semiconductor Sensor for Ozone (2022). https://www.allelectronics.com/mas_assets/media/ allelectronics2018/spec/MQ-131.pdf (accessed Nov. 12, 2022)
- Technical Data MQ-135 Gas Sensor (2022). https://www.olimex.com/Products/Components/Sensors/ Gas/SNS-MQ135/resources/SNS-MQ135.pdf (accessed Nov. 12, 2022)
- Technical Data MQ-7 Gas Sensor (2022). http://edge.rit.edu/edge/R13401/public/FinalDocuments/ Monitor/Appendix B Sensors.pdf (accessed Nov. 12, 2022)
- DHT 11 Humidity & Temperature Sensor (2022). https://osoyoo.com/driver/DHT11-datasheet.pdf (accessed Nov.12, 2022)
- Toma, C., Alexandru, A., Popa, M., & Zamfiroiu, A. (Aug. 2019). IoT Solution for Smart Cities' Pollution Monitoring and the Security Challenges. *Sensors*, 19(15), 3401. https://doi.org/10.3390/S19153401.
- Kiruthika, R., & Umamakeswari, A. (2017). Low cost pollution control and air quality monitoring system using Raspberry Pi for Internet of Things, in *International Conference on Energy, Communication, Data Analytics and Soft Computing, ICECDS 2017*, Jun. 2018, pp. 2319–2326. https://doi.org/10.1109/ICECDS.2017.8389867.
- Srivastava, H., Mishra, S., Das, S. K., & Sarkar, S. (2020). "An IoT-Based Pollution Monitoring System Using Data Analytics Approach," in Lecture Notes in Electrical Engineering, 686,187–198, doi: https:// doi.org/10.1007/978-981-15-7031-5_18.
- Okokpujie, K., Noma-Osaghae, E., Modupe, O., John, S., & Oluwatosin, O. (2018). A smart air pollution monitoring system. *Int J Civ Eng Technol*, 9(9), 799–809.

- Gupta, H., Bhardwaj, D., Agrawal, H., Tikkiwal, V. A., & Kumar, A. (2019). An IoT Based Air Pollution Monitoring System for Smart Cities, in *1st IEEE International Conference on Sustainable Energy Tech*nologies and Systems, ICSETS Feb. 2019, pp. 173–177. https://doi.org/10.1109/ICSETS.2019.8744949.
- Esfahani, S., Rollins, P., Specht, J. P., Cole, M., & Gardner, J. W. (2020). Smart City Battery Operated IoT Based Indoor Air Quality Monitoring System, in *Proceedings of IEEE Sensors*, Oct. vol. 2020-Oct. https://doi.org/10.1109/SENSORS47125.2020.9278913.
- Dur, T. H., Arcucci, R., Mottet, L., Solana, M. M., Pain, C., & Guo, Y. K. (Apr. 2020). Weak constraint gaussian processes for optimal sensor placement. *Journal of Computer Science*, 42, 101110. https://doi. org/10.1016/J.JOCS.2020.101110.
- Krause, A., Singh, A., & Guestrin, C. (2008). Near-Optimal Sensor Placements in Gaussian processes: theory, efficient algorithms and empirical studies. J Mach Learn Res, 9, 235–284.
- Longi, K. et al. (2020) Sensor Placement for Spatial Gaussian Processes with Integral Observations. Proceedings of Machine Learning Research 124, 1009-1018
- Tajnafoi, G. (2021). Variational Gaussian Process for Optimal Sensor Placement, *Appl. Math66(2)*, 287–317. https://doi.org/10.21136/AM.2021.0307-19.
- Singh, S., Singh, P., Garg, R., & Mishra, P. K. (2015) Big Data: Technologies, Trends and Applications, Int. J. Comput. Sci. Inf. Technol., 6(5), 4633–4639, Accessed: May 04, 2021. [Online]. Available: www. ijcsit.com.
- Hrehova, S. (2018). Brief overview of the concept of big data. Dec. https://doi.org/10.4108/ eai.6-11-2018.2279366.
- The Origins of Big Data KDnuggets (accessed May 04, 2021). https://www.kdnuggets.com/2017/02/ origins-big-data.html
- Ward, J. S., & Barker, A. (2013). Undefined By Data: A Survey of Big Data Definitions, Accessed: May 04, 2021. [Online]. Available: http://bigdatawg.nist.gov/home.php.
- 56. Cartledge, C. (2016). "How Many Vs are there in Big Data?
- Raza, M. U., & Xujian, Z. (2020). A Comprehensive Overview of BIG DATA Technologies: A Survey, in ACM International Conference Proceeding Series, May pp. 23–31. https://doi. org/10.1145/3404687.3404694.
- Louridas, P., & Ebert, C. (Sep. 2016). Machine learning. *IEEE Software*, 33(5), 110–115. https://doi. org/10.1109/MS.2016.114.
- Talabis, M. R. M., McPherson, R., Miyamoto, I., Martin, J. L., & Kaye, D. (2015). Analytics defined. in Information Security Analytics (pp. 1–12). Elsevier.
- Ding, Z., Huang, Y., Yuan, H., & Dong, H. (Jan. 2020). Introduction to reinforcement learning. *Deep Reinf Learn Fundam Res Appl*, 47–123. https://doi.org/10.1007/978-981-15-4095-0_2/COVER.
- Kaelbling, L. P., Littman, M. L., & Moore, A. W. (1995). An introduction to reinforcement learning. Biol Technol Intell Auton Agents, 90–127. https://doi.org/10.1007/978-3-642-79629-6_5.
- 62. Hyndman, R. J., & G. Athanasopoulos (2021) Forecasting: Principles and Practice (3rd ed)
- Assimakopoulos, V., & Nikolopoulos, K. (2000). The theta model: a decomposition approach to forecasting, *Int. J. Forecast*, 16(4), 521–530, https://doi.org/10.1016/S0169-2070(00)00066-2.
- Rabiner, L. R., & Juang, B. H. (1986). An introduction to hidden Markov Models. *IEEE ASSP Mag*, 3(1), 4–16. https://doi.org/10.1109/MASSP.1986.1165342.
- Gómez-Losada, Á., Pires, J. C. M., & Pino-Mejías, R. (2018). Modelling background air pollution exposure in urban environments: Implications for epidemiological research, *Environ. Model. Softw*, 106, 13–21. https://doi.org/10.1016/J.ENVSOFT.2018.02.011.
- Bontempi, G., Ben Taieb, S., & Le Borgne, Y.A. (2012). Machine learning strategies for Time Series forecasting. Lect Notes Bus Inf Process, 138 LNBIP, 62–77. https://doi.org/10.1007/978-3-642-36318-4_3.
- Das, R., Middya, A. I., & Roy, S. (2021). High granular and short term time series forecasting of \$\hbox PM_{2.5} air pollutant - a comparative review, *Artif. Intell. Rev.* 55, 1253-1287. https://doi.org/10.1007/ S10462-021-09991-1.
- Calkoen, F., Luijendijk, A., Rivero, C. R., Kras, E., & Baart, F. (2021). Traditional vs. Machine-Learning Methods for Forecasting Sandy Shoreline Evolution Using Historic Satellite-Derived Shorelines, *Remote Sens.* 13(5), 934. https://doi.org/10.3390/RS13050934.
- Oancea, B., & Ciucu, Ş. C. (2014). Time series forecasting using neural networks, Proc. Int. Conf. APL Lang. its Appl. APL 1994, pp. 86–94, Jan.
- Wang, Y. (2017). A new concept using LSTM Neural Networks for dynamic system identification, in *Proceedings of the American Control Conference*, Jun. pp. 5324–5329, doi: https://doi.org/10.23919/ ACC.2017.7963782.
- Wang, J. Air quality data analysis and forecasting platform based on big data, in *Proceedings 2019 Chinese Automation Congress, CAC 2019*, Nov. 2019, pp. 2042–2046. https://doi.org/10.1109/CAC48633.2019.8996332.

- Nandini, K., & Fathima, G. (2019). Urban Air Quality Analysis and Prediction Using Machine Learning, in *1st International Conference on Advanced Technologies in Intelligent Control, Environment, Computing and Communication Engineering, ICATIECE*, pp. 98–102. https://doi.org/10.1109/ ICATIECE45860.2019.9063845.
- Mahalingam, U., Elangovan, K., Dobhal, H., Valliappa, C., Shrestha, S., & Kedam, G. (2019). A machine learning model for air quality prediction for smart cities, in *International Conference on Wireless Communications, Signal Processing and Networking, WiSPNET 2019*, pp. 452–457. https://doi. org/10.1109/WiSPNET45539.2019.9032734.
- Ma, J., et al. (Jan. 2020). Identification of high impact factors of air quality on a national scale using big data and machine learning techniques. *Journal Of Cleaner Production*, 244, 118955. https://doi. org/10.1016/j.jclepro.2019.118955.
- Doreswamy, K. S., Harishkumar, Y., Km, & Gad, I. (2020). Forecasting Air Pollution Particulate Matter (PM2.5) Using Machine Learning Regression Models, in *Procedia Computer Science*, 171, 2057–2066. https://doi.org/10.1016/j.procs.2020.04.221.
- Xayasouk, T., & Lee, H., (2018) Air Pollution Prediction System Using Deep Learning (Oct. 2018). WIT Trans. Ecol. Environ., 230,71–79. https://doi.org/10.2495/AIR180071.
- Barthwal, A., Acharya, D., & Lohani, D. (2021). Prediction and analysis of particulate matter (PM2.5 and PM10) concentrations using machine learning techniques, *J. Ambient Intell. Humaniz. Comput.* 2021, 1–16,. https://doi.org/10.1007/S12652-021-03051-W.
- Jeya, S., & Sankari, L. (2020). Air Pollution Prediction by Deep Learning Model, Proc. Int. Conf. Intell. Comput. Control Syst. ICICCS 2020, pp. 736–741, May. https://doi.org/10.1109/ICI CCS48265.2020.9120932.
- Franceschi, F., Cobo, M., & Figueredo, M. (2018). Discovering relationships and forecasting PM10 and PM2.5 concentrations in Bogotá, Colombia, using Artificial Neural Networks, Principal Component Analysis, and k-means clustering, *Atmos. Pollut. Res*, 9(5), 912–922. https://doi.org/10.1016/J. APR.2018.02.006.
- Chang, Y. S., Chiao, H. T., Abimannan, S., Huang, Y. P., Tsai, Y. T., & Lin, K. M. (Aug. 2020). An LSTM-based aggregated model for air pollution forecasting. *Atmos Pollut Res*, 11(8), 1451–1463. https://doi.org/10.1016/J.APR.2020.05.015.
- Moursi, A. S., El-Fishawy, N., Djahel, S., & Shouman, M. A. (2021). An IoT enabled system for enhanced air quality monitoring and prediction on the edge, *Complex Intell. Syst1*, 1–25. https://doi. org/10.1007/S40747-021-00476-W.
- 82. Diaconescu, E., The use of NARX Neural Networks to predict Chaotic Time Series.
- Ketu, S., & Mishra, P. K. (2021). Scalable kernel-based SVM classification algorithm on imbalance air quality data for proficient healthcare, *Complex Intell. Syst1*, 1–19. https://doi.org/10.1007/ S40747-021-00435-5.
- 84. BF, W. (1995). The Chi square test: an introduction, COMSIG Rev, 4(3), 61-64
- Zhong, H., Yin, C., Wu, X., Luo, J., & He, J. (2020). AirRL: a reinforcement Learning Approach to Urban Air Quality Inference. *Mar.* https://doi.org/10.48550/arxiv.2003.12205.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Springer Nature or its licensor (e.g. a society or other partner) holds exclusive rights to this article under a publishing agreement with the author(s) or other rightsholder(s); author self-archiving of the accepted manuscript version of this article is solely governed by the terms of such publishing agreement and applicable law.



Amisha Gangwar is currently pursuing a MS degree in Applied Data Science from the School of Information Studies, Syracuse University, New York, US. She received her B. Tech. degree in Computer Science & Engineering from the Department of Electronics and Communication (J. K. Institue of Applied Physics and Technology), University of Allahabad, India in 2022. Her research interests include Machine Learning, Data Science, Internet of Things, and Computer Vision Deep Learning. She has published three research papers in the proceedings of peer-reviewed international conferences. She did various internships and training programs in reputed organizations.



Sudhakar Singh received his Master's and Ph.D. degree from Department of Computer Science, Institute of Science, Banaras Hindu University, India in 2010 and 2016 respectively. He is currently an Assistant Professor (Computer Science) in the Department of Electronics and Communication at the University of Allahabad, India. His current research interest includes Big Data Analytics, High Performance Computing, Machine Learning, Internet of Things, and Wireless Networks. He has published several research papers in referred international journals and proceedings of peer-reviewed international conferences. He is the reviewer and technical program committee member of various international journals and conferences. He is also the member of various professional and scientific societies.

Richa Mishra received her M.Sc. in Computer Science from Banaras Hindu University, Varanasi, India, in 2012 and her Ph.D. in Computer Science and Engineering from Pondicherry University, Puducherry, India, in 2016. She joined as an Academic Associate in Information Technology, Indian Institute of Management, Kashipur, India in 2017. From 2017 to 2018, she worked for CHRIST (Deemed to be University), Bangalore, and presently working as an Assistant Professor (Computer Science) in Department of Electronics and Communication at the University of Allahabad, Prayagraj, India. Her current research interest includes Computer Vision, Artificial Intelligence, Machine Learning, and Internet of Things. She has published several research papers in peer reviewed journals and international conference proceedings. She is a reviewer of several reputed journals.



Shiv Prakash received his Ph.D. degree and M.Tech. degree in Computer Science and Technology from Jawaharlal Nehru University, New Delhi in 2014 and 2010 respectively. His current research interest focuses on the Internet of Connected Vehicles (IoV), Electronic Vehicles (EV), Artificial Intelligence, Big Data Analytics, Biometric Security, Cloud Computing, Computer Networks, Machine Learning , Network Security, and IoT use cases of Sensor Networks. He has an excellent educational record throughout. He is a Reviewer of several referred International journals of repute (including ACM, IEEE, Taylor and Francis, Elsevier, Springer, Wiley, etc.) and published 40+ research papers in various peer-reviewed International Journals and Transactions (including IEEE, Taylor and Francis, Elsevier, Springer, Wiley, American Scientific Publishers, etc.) and around 20+ research papers in proceedings of various peer-reviewed conferences in India and abroad. He has published several papers in journals having very high ISI impact factors. He is a member of various International Asso-

ciations, Societies, and Scientific Committees. He is engaged in research in the collaboration with National and International organizations of repute.